Author’s response to reviews

Title: Lessons from a Training Needs Assessment to strengthen the capacity of Routine Immunization Service Providers in Nigeria

Authors:
Linda Arogundade (lindaomosefe@yahoo.com; linda.arogundade@dclnigeria.com)
Titilola Akinwumi (titi.akinwumi@dclnigeria.com; titimakay@gmail.com)
Ebubechi Nwaononiuwu (Ebubechi.Nwaononiuwu@gmail.com; Ebubechi.Nwaononiuwu@dclnigeria.com)
Joshua Ezika (joshykime@gmail.com; joshua.ezika@dclnigeria.com)
Inuwa Yau (inuwabarau@yahoo.com)
Shola Molemodile (sholamole@gmail.com; shola.mole@dclnigeria.com)
Chizoba Wonodi (chizobabw@gmail.com)

Version: 4 Date: 18 Jul 2019

Author’s response to reviews:

Editor Comments and Author Responses

-- Thank you for carrying out the requested revisions. Please may you also state the full name of the NHREC ethics committee, and the full name of each health institutions ethics committee in your 'Ethics approval and consent to participate section'.

Response: Done. Please see 'Ethics approval and consent to participate section' for a detailed description of the name of each committee that was influential in granting approval for conducting the study. The process was a 3 part thus beginning from the National level to the states and then to the local offices.